Author's response to reviews

Title: Vertebral artery dissection presenting as a Brown-Sequard syndrome: a case report

Authors:

Saul Miller (millersaul@gmail.com)
Dan Kottachchi (dankott@gmail.com)
Eli Miller (elimill@gmail.com)

Version: 4 Date: 1 November 2009

Author's response to reviews:

Dear BioMed Central Editorial Production Team

Thank you for your comments

1) I now include the ethnic origin and nationality of the patient in paragraph 2 of the case presentation "He was born in the Philippines and had lived in Canada since early childhood."

2) I now cite the figures in paragraph 5 of the case presentation. I have uploaded higher quality figures. I removed the original figure 2 as I was not satisfied with the image quality and could not prepare a substantially higher quality image from the original MR scans. I did not think that removing this figure detracted significantly from the paper.

Thanks

Saul Miller